Author’s response to reviews

Title: Risk Factors for Mycobacterium ulcerans Infection (Buruli Ulcer) in Togo — A Case-Control Study in Zio and Yoto Districts of the Maritime Region

Authors:

Issaka Maman (mamanissaka@yahoo.fr)

Tchadjobo Tchacondo (tchacondotchadjobo@yahoo.fr)

Abiba Banla Kere (kerebanla@yahoo.fr)

Ebekalisai Piten (pitenubtogo@yahoo.fr)

Marcus Beisner (beissner@lrz.uni-muenchen.de)

Yiragnima Kobara (kobarabasile@gmail.com)

Komlan Kossi (kossik2003@yahoo.fr)

Kossi Badziklou (pnlthatogo@yahoo.fr)

Franz Xaver Wiedemann (franz.wiedemann@dahwtogo.org)

Komi Amekuse (komi.amekuse@dahwtogo.org)

Gisela Bretzel (bretzel@lrz.uni-muenchen.de)

Damintoti Simplice Karou (simplicekarou@hotmail.com)

Version: 2 Date: 06 Nov 2017

Author’s response to reviews:

COVER LETTER FOR SUBMISSION OF REVISION MANUSCRIPT

04.11.2017

Dear Editor,

We hereby submit a revision of our manuscript entitled “Risk Factors for Mycobacterium ulcerans Infection (Buruli Ulcer) in Togo — A Case-Control Study in Zio and Yoto Districts of
the Maritime Region” by Issaka Maman as corresponding author to be considering for publication as Research article in the BMC Infectious Diseases.

With reference to the report of the editor revision of the manuscript after the second version submission of this article, we have formatted our manuscript with little changes to conform to the journal guidelines, which are presented below point-by-point:

1. Availability of data and materials section

Line 1, Page 16: We have reviewed the statement regarding the Availability of data and materials section in the text. We confirm that “All data generated or analysed during this study are included in this published article [and its supplementary information files]”. We therefore submit the questionnaire form used for data collection as supplementary form.

2. Authors’ Contributions section

Line 1, Page 17: We change the initials of the author Damintoti Simplice Karou to DSK

3. Submission of the figure

The Figure 1 has been re-uploaded in the submission of the revised article.

4. Submission of the manuscript

The manuscript has been uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables also have been submitted as clean versions.
Issaka Maman, MSc. (On behalf of all authors)

National Reference Laboratory for Buruli ulcer Manager

Institut National d’Hygiène

Lomé, Togo